Author's response to reviews

Title: Age, gender and disability predict future disability in older people: the Rotterdam Study

Authors:

Umit Tas (u.tas@erasusmc.nl)
Ewout W Steyerberg (e.steyerberg@erasusmc.nl)
Sita MA Bierma-Zeinstra (s.bierma-zeinstra@erasusmc.nl)
Albert Hofman (a.hofman@erasusmc.nl)
Bart W Koes (b.koes@erasusmc.nl)
Arianne P Verhagen (a.verhagen@erasusmc.nl)

Version: 4 Date: 26 April 2011

Author's response to reviews:

Dear Editor,

Hopefully we now revised the manuscript according to all the comments.
Yours, Arianne